Author's response to reviews

Title: Sequential introduction of single room isolation and hand hygiene campaign in the control of methicillin-resistant Staphylococcus aureus in intensive care unit

Authors:

Vincent CC Cheng (vcccheng@hkucc.hku.hk)
Josepha WM Tai (taiwm@ha.org.hk)
WM Chan (chanwm4@ha.org.hk)
Eric HY Lau (ehylau@hku.hk)
Jasper FW Chan (jaspchan@gmail.com)
Kelvin KW To (kelvinto@hkucc.hku.hk)
Iris WS Li (dririsli@yahoo.com.hk)
PL Ho (plho@hkucc.hku.hk)
KY Yuen (kyyuen@hkucc.hku.hk)

Version: 5 Date: 1 September 2010

Author's response to reviews: see over
Dear Dr Melissa Norton, MD
Editor-in-Chief
BMC-series journals

Re: Sequential introduction of single room isolation and hand hygiene campaign in the control of methicillin-resistant *Staphylococcus aureus* in intensive care unit

Thank you for revising our manuscript. We have already followed the suggestion and the amendment is listed in the point to point reply below.

Thank you for your kind attention.

Dr. Kwok-Yung Yuen
Department of Microbiology
The University of Hong Kong
Corresponding author
Major revisions (we require the author to make these changes)

Affiliations: Each affiliation should be written in full, in the following format: Department, Institute, (Street), City, (Postal code), Country. This may mean repeating institutional addresses if the authors are from different departments within an institution.

**Ans: It is amended in the final version.**

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above

Track changes: please remove

**Ans: It is removed.**

Highlighting/coloured text: please remove.

**Ans: It is removed.**

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

**Ans: Typographic error has been corrected.**